Author’s response to reviews

Title: Do endometrial lesions require removal? A retrospective study

Authors:

Ting Jiang (cherryj2013@hotmail.com)
Qing Yuan (yuanqing@51mch.com)
Qin Zhou (qzhou@shmu.edu.cn)
Yiping Zhu (zhuyiping@51mch.com)
Siji Lv (lvsiji@51mch.com)
Yanling Cao (caoyanling@51mch.com)
Qin Wang (13122557949@163.com)
Kunming Li (likunming@hotmail.com)
Dong Zhao (hendryz@gmail.com)

Version: 4 Date: 24 Apr 2019

Author’s response to reviews:

Dear Editor:

We had made some formatting changes according to your requests.

1. Availability of Data and Materials statement: We choose the datasets used and/or analysed during the current study available from the corresponding author on reasonable request.

2. Ethics approval: We added 'The study protocol was approved by the Ethical Committee of the Tongji University School of Medicine' to the methods section and declarations.

3. Consent to participate: Written informed consent was obtained from all participants.

4. Remove files: We remove them from the submission.

5. Clean manuscript: We uploaded our manuscript, tables, and figures as single, final, clean version, and Figures remain uploaded as separate files.
Sincerely,

Dong Zhao M.D.